\documentclass[11pt]{article}
 \setlength{\textheight}{10.4in}
 \setlength{\topmargin}{-1in}
 \setlength{\headheight}{0in}
 \setlength{\topsep}{0in}
 \setlength{\marginparwidth}{0in}
 \setlength{\marginparsep}{0in}
% \setlength{\footskip}{4pc}
%\setlength{\footheight}{0in}
 \setlength{\oddsidemargin}{0in}
 \setlength{\evensidemargin}{0in}
 \setlength{\textwidth}{6.4in}
 \setlength{\parskip}{4pt}
 \setlength{\parindent}{0pt}
\parindent=0in
\parskip=1pc
\begin{document}
\thispagestyle{empty}
{\large\bf Quizzes} are take-home problem sets, one per week. Each quiz
has a corresponding sample quiz and solutions. It is assumed that you
will read the sample quiz and solutions, in order to understand the
background material from mathematics, physics, science and engineering.

Take-home exams are an advantage if you are traveling or have a full
work schedule. The 14 quizzes will be produced on the fly, as the
semester progresses.

Quizzes are assigned weekly, and due the next week. If it is week 5 and
this message appears instead of Quiz5, then there is a mistake.
Sometimes, the missing Quiz is due to browser caching. Please reload the
PDF for the missing quiz, to insure there really is a mistake. Ask a
friend with a different computer to confirm the mistake. Then
please call, write email or ask questions in person.

\end{document}
